**Author's response to reviews**

**Title:** Combination of four molecular markers improves thyroid cancer cytologic diagnosis and patients management

**Authors:**

Federica Panebianco (panebiancof@upmc.edu)
Chiara Mazzanti (c.mazzanti@fondazionepisascienza.org)
Sara Tomei (stomei@sidra.org)
Paolo Aretini (p.aretini@fondazionepisascienza.org)
Sara Franceschi (sara.franceschi@fondazionepisascienza.org)
Francesca Lessi (f.lessi@fondazionepisascienza.org)
Giancarlo Di Coscio (g.dicoscio@ao-pisa.toscana.it)
Generoso Bevilacqua (generoso.bevilacqua@med.unipi.it)
Ivo Marchetti (i.marchetti@ao-pisa.toscana.it)

**Version:** 2  
**Date:** 22 April 2015

**Author's response to reviews:**

Dear Editor,

As your journal required, the following revisions have been done:

- The consent statement has been added in the section Materials and Methods, sub-section Ethical board
- The acknowledgements have been added in a dedicated section
- We reviewed your policy on data to be submitted publicly, but our study did not generate data of this type.

Sincerely

Federica Panebianco